Author's response to reviews

Title: Minimally invasive and computer-navigated total hip arthroplasty. A qualitative and systematic review of the literature.

Authors:

Inge H.F. Reininga (i.reininga@orth.umcg.nl)
Wiebren Zijlstra (w.zijlstra@med.umcg.nl)
Robert Wagenmakers (r.wagenmakers@orth.umcg.nl)
Alexander L. Boerboom (a.l.boerboom@orth.umcg.nl)
Bregtje P. Huijbers (bregtjehuijbers@hotmail.com)
Johan W. Groothoff (j.w.groothoff@med.umcg.nl)
Sjoerd K. Bulstra (s.k.bulstra@orth.umcg.nl)
Martin Stevens (m.stevens@orth.umcg.nl)

Version: 3 Date: 11 May 2010

Author's response to reviews:

Major revisions:
- A section titled "Additional files" has been added at the end of the manuscript.
- The heading of the introduction section has been changed into "Background".

Minor revisions:
- The title is now written in sentence case.
- All email addresses are included on the title page, according to the format of BMC musculoskeletal disorders
- Headings in the manuscript that were underlined, are now written in Italic.
- Figure 1 has been cropped, so that little white space around the figure remained.